Author’s response to reviews

Title: EFFECTS OF WHOLE-BODY ELECTROMYOSTIMULATION ON HEALTH AND PERFORMANCE: SYSTEMATIC REVIEW

Authors:

Alvaro De Pano Rodríguez (alvarodepano@gmail.com)

Jose Vicente Beltran-Garrido (jose.vicente@euseste.es)

Vicenç Hernández González (vicens_h_g@didesp.udl.cat)

Joaquim Reverter Masia (reverter@didesp.udl.cat)

Version: 2 Date: 07 Mar 2019

Author’s response to reviews:

We are very grateful for the contributions received by the reviewer:

1.- Line 58: we have changed cm² to cm³

2.- Line 78: we have included in the background some information about the benefits of electrostimulation as an effective treatment of spasticity in subjects with neurological disorders.

3.- Tables 3 and 4: we have updated the duty cycle information

4.- Tables 3 and 4 we have changed the term "cronaxie" to "on time of pulse"

5.- Tables 3 and 4: we have included the data "rise and decay time" when it was provided by the authors.

6.- We have uploaded our manuscript as a single, final, clean version.